Author's response to reviews

Title: Effectiveness of a new health care organization model in primary care for chronic cardiovascular disease patients based on a multifactorial intervention: The PROPRESE randomized controlled trial.

Authors:

Domingo Orozco-Beltran (dorozcobeltran@gmail.com)
Esther Ruescas-Escolano (eshterre@hotmail.com)
Ana I Navarro-Palazón (anaisabel.napa@gmail.com)
Alberto Cordero (acorderofort@gmail.com)
María Gaubert-Tortosa (maga_4u@hotmail.com)
Jorge Navarro-Perez (navarro_jorper@gva.es)
Concepcion Carratala-Munuera (atencion.primaria@umh.es)
Salvador Pertusa-Martínez (spertusam@medynet.com)
Enrique Soler-Bahilo (soler_enr@gva.es)
Francisco Brotons-Muntó (brotons_fra@gva.es)
Jose Bort-Cubero (bort_jos@gva.es)
Miguel A Nuñez-Martinez (nunyez_mig@gva.es)
Vicente Bertomeu-Martinez (vbertomeu@secardiologia.es)
Vicente F Gil-Guillen (vte.gil@gva.es)

Version: 2 Date: 18 April 2013

Author's response to reviews:

Dear editorial board of BMC Health Services Research,

As you requested, we have added the Trial Registration Number at the end of the abstract in the submission system and we have sent to you ethical approval and external funding proofs. In addition, we have uploaded the manuscript again because we noticed a mistake.

Sincerely yours,

Domingo Orozco-Beltran on behalf of the authors.